Author's response to reviews

Title: Dysphagia as a manifestation of Oesophageal Tuberculosis: a report of two cases.

Authors:

Joana Gomes (jlobogomes@gmail.com)
Ana Antunes (anaoliveirinha2@gmail.com)
Aurora Carvalho (dra.aurora.carvalho@gmail.com)
Raquel Duarte (raquelafduarte@gmail.com)

Version: 7 Date: 19 July 2011

Author's response to reviews: see over
Dear Editor of Journal of Medical Case Reports,

We enclose here with the revised manuscript entitled: “Dysphagia as a manifestation of Oesophageal Tuberculosis: a report of two cases.” previously submitted to Journal of Medical Case Reports.

We have made the formatting changes suggested by the Associate Editor – the changes are in red.

- we restructured the authors’ contribution section;
- we remove any text from figures which may reveal the patient’s identity;
- we included the ethnicity, sex and age of the patients in the abstract.

All the authors read and approved the revised manuscript.

We remain with our kindest regards.

Yours sincerely,

Joana Gomes, MD